Author’s response to reviews

Title: Occupational burnout and empathy influence blood pressure control in primary care physicians

Authors:
Oriol Yuguero (oriol.yuguero@gmail.com)
Josep Ramon Marsal (jmarsal lleida.ics@gencat.cat)
Montserrat Esquerda (mesquerda@comll.cat)
Jorge Soler-González (jorgesolergonzalez@gmail.com)

Version: 1 Date: 29 Nov 2016

Author’s response to reviews:

Distinguished Editor,

Thanks for your reply. First of all, excuse the error of the linguistic revision of the article. The attached manuscript was incorrect, since it was the previous one to the revision realized by a professional translator. Please excuse the inconvenience.

In second place we have adapted the Abstract as recommended.

Finally we have adapted the section of Declarations as it comes in the guide for publication. We have also revised the bibliography to suit your journal's standards.

Sincerely, and thank you for your attention.

a greeting

Oriol Yuguero